Author's response to reviews

**Title:** Prevalence and Correlation between Depression, Anxiety, and Other Features in Outpatients with COPD in China

**Authors:**

- Peian Lou (loupeian2004@yahoo.com.cn)
- Yanan Zhu (yanan01112@163.com)
- Peipei Chen (71772690@qq.com)
- Pan Zhang (xzzhangpan@126.com)
- Jiaxi Yu (zxyjx007@163.com)
- Ning Zhang (494174@qq.com)
- Na Chen (147819560@qq.com)
- Lei Zhang (18905208811@189.cn)
- Hongmin Wu (wuhongmin19870512@163.com)
- Jing Zhao (354551183@qq.com)

**Version:** 3  **Date:** 6 July 2012

**Author's response to reviews:** see over
Dear editor:

According to your requirements, I have revised my manuscript as fallow:

1、Language has been the mother tongue by edanz group.

2、How were the patients and control sampled? It had been explained in Sampling strategy and the first paragraph of results.

3、The discussion has been revised.

Sincerely yours,

Peian L. on behalf of the authors.